Author’s response to reviews

Title: p53 expression in patients with ulcerative colitis - associated with dysplasia and carcinoma: a systematic meta-analysis

Authors:
Xiaohong Lu (luxiaohong_2017@163.com)
Yuanjie Yu (kpo0336644@sina.com)
Shiyun Tan (dtuo8852076@sina.com)

Version: 4 Date: 30 Aug 2017

Author’s response to reviews:

Editor Comments:

1. Please rename the "Introduction" section to "Background," and "Materials and methods" to "Methods" in order for the manuscript format to meet the journal guidelines.
Response: We have renamed the "Introduction" section to "Background," and "Materials and methods" to "Methods".

2. Please move the Abbreviations section to the end of the paper, before the Declarations section.
Response: We have moved the Abbreviations section to the end of the paper, before the Declarations section.

3. Please specify who was responsible for writing the article and revising it for important intellectual content in the Author Contributions section.
Response: We have specified LXH was responsible for writing the article and revising it for important intellectual content in the Author Contributions section.

4. Please refer to table S1 in the main text of the manuscript.
Response: We have referred to table S1 in the main text of the manuscript.